Author's response to reviews

Title: A prospective, multi-method, multi-disciplinary, multi-level, collaborative, social-organisational design for researching health sector accreditation [LP0560737]

Authors:

Jeffrey Braithwaite (j.braithwaite@unsw.edu.au)
Johanna I Westbrook (j.westbrook@unsw.edu.au)
Marjorie Pawsey (mpawsey@achs.org.au)
David Greenfield (d.greenfield@unsw.edu.au)
Justine Naylor (Justine.Naylor@sswhs.nsw.gov.au)
Rick A Iedema (r.iedema@unsw.edu.au)
Bill Runciman (wrunciman@bigpond.com)
Sally Redman (Sally.Redman@saxinstitute.org.au)
Christine Jorm (cjorm@bigpond.net.au)
Maureen Robinson (Maureen.Robinson@communio.com.au)
Sally Nathan (s.nathan@unsw.edu.au)
Robert Gibberd (Robert.Gibberd@newcastle.edu.au)

Version: 2 Date: 11 September 2006

Author's response to reviews:

Dear BMC Editorial team

All issues have now been dealt with including ethics approval mentioned, separate figure files, cropping of figures and elimination of figure placement text.

Best wishes
Jeffrey